Author's response to reviews

Title: Tight correlation between expression of the Forkhead transcription factor FOXM1 and HER2 in human breast cancer

Authors:

Nuran Bektas (nbektas@ukaachen.de)
Anette ten Haaf (atenhaaf@ukaachen.de)
Juergen Veeck (juergen.veeck@rwth-aachen.de)
Peter Johannes Wild (peter.wild@usz.ch)
Juliane Luescher-Firzlaff (jluescher-firzlaff@ukaachen.de)
Arndt Hartmann (arndt.hartmann@klinik.uni-regensburg.de)
Ruth Knuechel (rknuechel-clarke@ukaachen.de)
Edgar Dahl (edahl@ukaachen.de)

Version: 3 Date: 5 February 2008

Author's response to reviews:

Dear BioMed Central Editorial Production Team,

we made the following formatting changes requested:

1. We deleted the key words from the title page.
2. We indicated in the authors' contribution section that all authors read and approved the final manuscript.
4. The references section has been included in the main text.

Thank you very much,

Yours sincerely,

Edgar Dahl